Author's response to reviews

Title: Encephalitis temporally associated with Japanese encephalitis vaccine: Case report

Authors:

Na Jia (jiana79_41@hotmail.com)
Qiu Min Zhao (prettynana@sohu.com)
Xiao Fang Guo (prettynana@sohu.com)
Jing Xia Cheng (prettynana@sohu.com)
Chao Wu (prettynana@sohu.com)
Shu Qing Zuo (prettynana@sohu.com)
Pei Fang Dai (prettynana@sohu.com)
Jun Ying Zhao (prettynana@sohu.com)
Jiu Song Zhang (zhang_jiusong@yahoo.com.cn)

Version: 3 Date: 2 December 2010

Author's response to reviews:

Dear Editor,

As your requests, we provided copies of the informed consent documents by sending them via email to editorial@biomedcentral.com. We revised the Informed consent section to clarify who the consent was obtained from and move this section into the Acknowledgements section. In addition, we added an ethical statement within the Methods section and clarify the name of the Ethical Committee. The four cases were confirmed for Japanese encephalitis infection by patients’ cerebrospinal fluid and/or serum samples detection, which were described within the Method section.

Best regards,

Jiu-Song Zhang